Author's response to reviews

Title: The 3-hydroxy-3-methylglutaryl-coenzyme A reductase inhibitors, simvastatin, lovastatin and mevastatin inhibit proliferation and invasion of melanoma cells.

Authors:

Sharon A Glynn (glynnsh@mail.nih.gov)
Dermot O'Sullivan (dermot.osullivan2@mail.dcu.ie)
Alex J Eustace (alex.eustace@dcu.ie)
Martin Clynes (martin.clynes@dcu.ie)
Norma O'Donovan (norma.odonovan@dcu.ie)

Version: 6 Date: 8 January 2008

Author's response to reviews:

Dear Editor

All of the formatting changes outlined below have been completed and the manuscript has been thoroughly proofed.

1. 'Title:' removed from the top of the title page.
2. Keywords removed
3. Graphics in the legend of Figure 5 have been removed and placed on the figure.
4. All figures have been formatted as outlined in the instructions
5. The reference to "Table 2 contains results of student's t-test" - I do not see a table 2? has been removed. This was carried over from revision 1 but the p values are now inserted in the figure so the extra table is no longer necessary.
6. All references to the "Student's t-test" have been corrected.

Yours sincerely
Norma O'Donovan and Sharon Glynn